Author's response to reviews

Title: Physical activity in Iranian older adults who experienced fall during the last 12 months

Authors:

Leili Salehi (leilisalehi83@yahoo.com)
Behjat Shokrvash (shokrvash@tbzmed.ac.ir)
Ensiyeh Jamshidi (ensiyeh_jamshidi@yahoo.com)
Ali Montazeri (montazeri@acecr.ac.ir)

Version: 5  Date: 27 September 2014

Author's response to reviews: see over
Dear Dr. Helbostad,

MS: 1405054127131351
Physical activity in Iranian older adults who experienced fall during the last 12 months

Thank you for your kind e-mail. Please find the following point-by-point responses as requested:

Editorial Comments:

"Thank you for the second revision of the manuscript. I have two minor comments to the last version that I want to pay attention to:
Line 1, page 6: "We used this information to aid in item writing": the sentence is not clear, please rewrite
Since later we say the same thus to prevent confusion we deleted this sentence
Page 10: limitations, line 15-16: You need an argument for why your sample has a high socio economic status!!
We revised the sentence with an argument to comply with the recommendation. We feel now this revised sentence goes even better with our previous statement on family cohesiveness, and social support:

These elderly might differed from others in terms of family cohesiveness, and social support. We did not collect data for these, but given that most participants were married and living with someone, we speculate that our sample had better conditions than the general older population.

Otherwise you have replied well to the concerns raised in my previous feedback, and I am therefore pleased to accept the manuscript for publication in BMC Geriatrics."

Thank you for your time and efforts to improve the manuscript.

Hope you find the corrections satisfactory.
I wish you all the best.
Kind regards
Ali Montazeri